Author's response to reviews

Title: Stay@Work: Participatory Ergonomics to prevent low back and neck pain among workers: design of a randomised controlled trial to evaluate the (cost-)effectiveness

Authors:

Maurice T Driessen (m.driessen@vumc.nl)
Johannes R Anema (j.anema@vumc.nl)
Karin I Proper (k.i.proper@vumc.nl)
Paulien M Bongers (paulien.bongers@tno.nl)
Allard J van der Beek (a.vanderbeek@vumc.nl)

Version: 6 Date: 28 October 2008

Author's response to reviews:

I made 3 changes in the reference list. In references nr: 70, 65 and 30 I wrote et al. Now, according to the BMC format, I wrote down all authors' names instead of using et al.